CONSENSUS STATEMENT



Transition from pediatric to adult care in patients with Turner syndrome in Italy: a consensus statement by the TRAMITI project

T. Aversa^{1,2} · L. De Sanctis^{3,4} · M. F. Faienza⁵ · A. Gambineri⁶ · A. Balducci⁷ · R. D'Aprile^{8,9} · C. Di Somma¹⁰ · C. Giavoli¹¹ · A. Grossi¹² · M. C. Meriggiola¹³ · E. Profka¹⁴ · M. Salerno¹⁵ · S. Stagi^{16,17} · E. Scarano¹⁸ · M. C. Zatelli¹⁹ · M. Wasniewska^{1,2}

Received: 4 October 2023 / Accepted: 11 January 2024 © The Author(s) 2024

Abstract

Purpose Transition from pediatric to adult care is associated with significant challenges in patients with Turner syndrome (TS). The objective of the TRansition Age Management In Turner syndrome in Italy (TRAMITI) project was to improve the care provided to patients with TS by harnessing the knowledge and expertise of various Italian centers through a Delphi-like consensus process.

Methods A panel of 15 physicians and 1 psychologist discussed 4 key domains: transition and referral, sexual and bone health and oncological risks, social and psychological aspects and systemic and metabolic disorders.

Results A total of 41 consensus statements were drafted. The transition from pediatric to adult care is a critical period for patients with TS, necessitating tailored approaches and early disclosure of the diagnosis to promote self-reliance and health-care autonomy. Fertility preservation and bone health strategies are recommended to mitigate long-term complications, and psychiatric evaluations are recommended to address the increased prevalence of anxiety and depression. The consensus also addresses the heightened risk of metabolic, cardiovascular and autoimmune disorders in patients with TS; regular screenings and interventions are advised to manage these conditions effectively. In addition, cardiac abnormalities, including aortic dissections, require regular monitoring and early surgical intervention if certain criteria are met.

Conclusions The TRAMITI consensus statement provides valuable insights and evidence-based recommendations to guide healthcare practitioners in delivering comprehensive and patient-centered care for patients with TS. By addressing the complex medical and psychosocial aspects of the condition, this consensus aims to enhance TS management and improve the overall well-being and long-term outcomes of these individuals.

Plain language summary

The TRansition Age Management in Turner syndrome in Italy (TRAMITI) project aims to improve care for individuals with Turner Syndrome (TS) during their transition from pediatric to adult care. A team of 15 physicians and 1 psychologist collaborated to create a comprehensive set of 41 consensus statements, covering four key areas: transition and referral, sexual and bone health and oncological risks, social and psychological aspects and systemic and metabolic disorders. The consensus statements highlight the importance of patient-centered care, early intervention and long-term monitoring. They emphasize a multidisciplinary approach to address the complex medical and psychosocial aspects of TS. During the critical transition period, tailored approaches and early disclosure of the diagnosis are recommended to promote self-reliance and healthcare autonomy. To mitigate long-term complications, the consensus addresses fertility preservation and bone health strategies. It also recommends psychological or psychiatric evaluations to tackle the increased prevalence of anxiety and depression in patients with TS. In addition, strategies for addressing the heightened risk of metabolic, cardiovascular and autoimmune disorders in patients with TS are proposed. Regular screenings and interventions are advised to effectively manage these conditions. Furthermore, cardiac abnormalities, including aortic dissections, require close monitoring and early surgical intervention if specific criteria are met. Overall, the TRAMITI consensus statement provides valuable insights and evidence-based recommendations. It offers guidance for healthcare practitioners in delivering comprehensive and patient-centered care

Extended author information available on the last page of the article

for individuals with TS. By addressing both medical and psychosocial aspects, the consensus aims to enhance TS management and improve the well-being and long-term outcomes of those affected by this genetic disorder.

Keywords Delphi consensus · Multidisciplinary care · Patient-centered approach · Transition management · Turner syndrome

Introduction

Turner syndrome (TS) is a genetic disorder characterized by the complete or partial loss of one of the two X chromosomes, affecting only women at an approximate rate of 1 in 2500 live births [1, 2]. The characteristic clinical features of TS include short stature, webbed neck, cardiac defects, endocrine and metabolic disorders and infertility [3, 4].

TS management poses unique challenges due to multisystem involvement, necessitating a coordinated multidisciplinary approach to ensure optimal patient care [5]. Regular monitoring and timely interventions across various medical specialties including cardiology, endocrinology and psychology are essential to mitigate the long-term complications of TS and improve quality of life (QoL) [5].

One of the most significant challenges in TS management is the transition from pediatric to adult care [6, 7]. This transition is a critical period as individuals with TS experience numerous physiological and psychosocial changes that require a shift in healthcare priorities [6]. Although a number of transition protocols have been proposed, many patients with TS do not receive appropriate care during this critical period, predisposing them to adverse health outcomes and threatening their QoL [8, 9].

Given the multifaceted challenges associated with TS, there is a need for comprehensive consensus about the transition of patients from pediatric to adult healthcare. The objective of the TRansition Age Management In Turner syndrome in Italy (TRAMITI) project was to improve the care provided to patients with TS at the national level by harnessing the experiences of various Italian centers through a Delphi-like consensus process.

Methods

The TRAMITI project employed a Delphi-like methodology. Members of the TRAMITI working group included 15 Italian physicians and 1 psychologist who are all experts in TS management, including both pediatric and adult endocrinologists, a cardiologist and a gynecologist. Members of the working group were selected based on their level of experience in their respective specializations (i.e., pediatric or adult), and in transitioning patients with TS from pediatric to adult care. The initial advisory board meeting for the project was convened in Rome on 21 December 2022. The primary objectives of this meeting were to establish the rationale and structure of the project, identify key topics and subtopics for consensus statements, delineate the roles and responsibilities of working subgroups and outline a methodology for these subgroups.

Prior to the meeting, a comprehensive literature search was conducted using Boolean operators and keywords relevant to various topics under consideration (Supplemental Methods; Online Resource 1). The results of this search were disseminated among the participants, ensuring that a foundational set of literature was readily available for reference throughout the project. Members of the advisory board could also suggest references.

Following extensive discussion, members of the advisory board agreed that the consensus statements should encompass four topics: transition and referral, sexual and bone development, systemic and metabolic disorders and social and psychological aspects (Supplemental Methods; Online Resource 1). Members also agreed to constitute four working subgroups in such a way as to ensure an equitable distribution of members across various specialties (Supplemental Methods; Online Resource 1). Each working subgroup was tasked with independently reviewing literature pertinent to their specific topic, developing shared insights and generating statements and arguments based on these insights. Working subgroups were advised to use the modified Grading of Recommendations, Assessment, Development, and Evaluation (GRADE) system [10, 11]. In the GRADE system, 1 indicates a strong recommendation and 2 indicates a weak recommendation. Quality of evidence is classified as +00 (low), ++0 (moderate) or +++ (high).

Following the initial meeting, each working subgroup held two asynchronous meetings. The first meeting, entitled 'Problem setting and analysis of the subtopic,' involved the review of the relevant literature, development of the subtopics and drafting of statements and supporting arguments. The first set of asynchronous meetings took place in February 2023. The second asynchronous meeting, entitled 'Definition of the statements for the subtopic,' encompassed discussion of the statements to clarify their wording, as well as any remaining points, and the finalization of the statements to be shared with the entire advisory board. The second set of asynchronous meetings took place in March 2023. Working subgroups were given the following criteria for drafting the statements: the statements should be in English, be affirmative, not contain double negatives, be concise, focused on a single theme and be supported by references.

The final meeting was held in Bologna on 27 April 2023. During this meeting, the work of each subgroup was consolidated, the statements and the arguments were reviewed collectively and, if necessary, refined. After that, members of the advisory board voted on each statement online selecting the options 'agree' or 'do not agree.'

Results

A total of 41 statements were drafted and subject to a vote (Table 1). All 16 members of the working group agreed with 39 statements. Fifteen members of the working group agreed with statement B1, while one member abstained. Similarly, 15 members of the working group agreed with statement D9 and one member disagreed.

Transition and referral

The transition process and its timing should be tailored to the unique needs of young adults with TS and should be based on interdisciplinary collaboration between pediatric and adult care specialists [5]. Ideally, transition should occur around the age of 18, as timely integration into an adult clinic is predictive of consistent and stable long-term followup (Statement A1) [12].

Early disclosure of TS diagnosis to parents and, subsequently, to the patient around the age of 11–13 years, promotes self-reliance and healthcare autonomy [5, 13]. Developmental age should take precedence over chronological age in determining communication timing (Statement A2) [14].

Since there is currently no validated TS-specific transition readiness assessment tool, the Transition Readiness Assessment Questionnaire (TRAQ) 5.0 could be recommended due to its well-defined structure and ease of use [15] (Statement A3). TRAQ is a validated instrument comprising 20 items designed to evaluate an individual's knowledge and self-assessed competencies in health-related skills [16].

In order to reduce the risk of chronic complications and comorbidities in women with TS, a timely initiation of hormone replacement therapy (HRT) for ovarian dysfunction, continued management of complications that began in childhood and proactive screening for comorbidities that may emerge in adulthood are necessary (Statement A4) [5, 17–19]. Adolescents with TS should be informed about how their condition is likely to affect their adult lives, about the risk of complications and the need for regular follow-up and preventive care (Statement A5) [5].

Identifying disease-specific knowledge is crucial due to the significance of self-efficacy in behavioral interventions [5, 20]. Before transitioning, a comprehensive review of TSrelated issues and future care plans is essential (Statement A6). Transition discussions should encompass the adolescent's progression into adulthood, including health, independence, education, career and social aspects [5, 7, 12, 17, 21, 22]. Psychological support, educational assistance and engagement with TS support groups are all invaluable in boosting self-esteem and essential skills [5, 7, 12, 17, 21, 22].

Telemedicine can be helpful in enhancing the sharing of information between patients and clinical staff, thus bolstering the effectiveness of consultations and adherence to follow-up appointments (Statement A7). Incorporating telemedicine platforms into transition strategies offers the potential for greater flexibility in scheduling visits and mitigating social and logistical hurdles for both patients and parents [13, 23, 24]. However, it is important to recognize that telemedicine should complement, rather than supplant, in-person visits.

Various models of transition have been proposed, ranging from vertically integrated care within a single hospital to care provided across different sites within the same healthcare system [7, 25]. Ideally, transition should take place in the same hospital where the patient received pediatric care (Statement A8). An essential element of transition is a medical record summary that includes TS-specific clinical information, which should be shared by the pediatric and adult endocrinologists with the patient and their caregivers (Statement A9) [26]. The Endocrine Society offers a specialized tool, the Clinical Summary and Transfer Record for Youth with Turner Syndrome, which can be utilized as a TS-specific transition aid [15].

Both pediatric and adult endocrinologists are regarded as the driving forces within the multidisciplinary care team [13]; the first transition visit should be managed by both specialists (Statement A10).

Sexual health and fertility

TS is usually accompanied by hypergonadotropic hypogonadism and amenorrhea [5, 27, 28]. As a result, spontaneous puberty occurs in 21–50% of TS girls and menarche in 15–30% of patients with mosaicism [5, 27, 28]. Thus, pubertal failure is frequent in girls with TS, and most need hormonal replacement therapy (HRT) for induction or completion of puberty, maintenance of secondary sex characteristics, attaining peak bone mass and normalizing uterine growth for potential pregnancies (Statement B1). To mimic physiological initiation and/or progression of puberty, treatment should begin at 11–12 years of age, or whenever pubertal development fails to progress [27].

Although the optimal regimen for estrogen replacement therapy (ERT) in girls with TS has not been established

Table 1 Final consensus statements

No	Statement	Rating
Working subgroup A: transition and referral		
A1	If possible, TS patients should be transitioned from pediatric to adult endocrinologist at the age of 18 years	1/++0
A2	The appropriate age could differ depending on the readiness of each patient	1/++0
A3	A validated assessment questionnaire should be used to evaluate transition readiness in TS patients	1/++0
A4	To ensure that patients with TS are physically fit for transition, milestones for the health management in adulthood should have been achieved	1/++0
A5	Adolescent patients with TS should be informed about all aspects of the syndrome during adult life, including the risk of complica- tions and the need for regular follow-up and preventive healthcare	1/++0
A6	Before transition, patients' and caregivers' knowledge about TS should be reviewed, including the need and willingness for any additional education, which should be personalized	1/++0
A7	During the transition process, telemedicine can enhance commu- nication and the sharing of information and increase flexibility in visit scheduling	2/+00
A8	Ideally, transition to adult care for TS patients should take place at the same hospital where the patient was treated by a pediatrician in the setting of a multidisciplinary team	1/++0
A9	A medical record summary containing essential TS-specific clinical information is necessary during transition from the care of pediat- ric to adult endocrinologist	1/++0
A10	The first transition visit should be managed by both pediatric and adult endocrinologists as an opportunity to ensure a continuity of care and to establish a comprehensive care plan	1/++0
Working subgroup B: sexual and bone health and oncological risk		
B1	Most girls with TS require HRT, initially for the induction of puberty and later for maintaining secondary sex characteristics, attaining peak bone mass and normalizing uterine growth for possible pregnancy	1/++0
B2	Transdermal preparations of ERT are the preferred regimen in girls with TS	1/+00
B3	Girls with TS typically have a normal uterus, and progestin/proges- terone must be added once breakthrough bleeding occurs or after 2 years of ERT	1/+00
B4	During transition, or before the transition process begins, individu- als with TS should be counseled that their ability to conceive spontaneously decreases rapidly with age and may not be present during adulthood and that spontaneous pregnancies are rare	1/+++
B5	Fertility assessment should be performed between the age of 12 and 14 years; the best procedure for preserving fertility in young women with TS and persistent ovarian function is oocyte or embryo cryopreservation	2/+00
B6	Osteopenia and osteoporosis are common features in young women with TS, with an estimated prevalence of 50% and an increased risk of early bone fractures	1/++0
B7	ERT is essential for the maintenance of bone health in TS, and early initiation is one of the most important determinants of bone health	1/+++
B8	Transdermal estrogens appear to have more beneficial effects on bone mass than oral preparations	2/+00

No	Statement	Rating
B9	Although overall oncologic risk in TS is similar or slightly higher compared with the general population, particular attention should be paid to the monitoring for specific tumors, such as thyroid carcinoma, meningioma and skin tumors	1/+00
B10	Individuals with a Y chromosome may be at risk of gonadoblas- toma, therefore gonadectomy should be discussed with these patients and their caregivers, taking into account the balance of the risks and benefits of the procedure	2/+00
Working subgroup C: social and psychological a.	spects	
C1	For an effective transition process, direct involvement of the patient with TS in the treatment path is essential	1/++0
C2	TS is associated with the presence of a characteristic neurocogni- tive profile that can influence the long-term adaptive and social functioning and the acquisition of full autonomy	1/++0
C3	TS is associated with psychiatric disorders, including a signifi- cantly elevated risk of anxiety and depression in adolescents and young adults	1/++0
C4	Anxiety and depression interfere with the psychological health and function of patients with TS	1/++0
C5	A periodic neuropsychological assessment during the age of development is needed to prevent the occurrence or worsening of psychopathological problems later in life	1/++0
C6	Psychodiagnostic assessment should be reinforced in adolescence and at the age of transition in patients with TS	1/++0
C7	Sexuality-related attitudes and behaviors of adolescents with TS should be discussed before the transition to adult care to ensure greater psycho-relational well-being	1/++0
C8	Adolescents and young women with TS have a lower QoL than the general population, with a marked impairment in psychosocial variables, although generally their socioeconomic status does not seem to be impaired	1/++0
С9	Since eating disorders are more frequent in adolescents with TS compared with the general population, this issue should be proac- tively investigated	1/++0
Working subgroup D: systemic and metabolic dis	orders	
DI	In young women, screening for DM using HbA1c and FPG is recommended annually, and with OGTT every two years, regard- less of BMI, family history, karyotype, previous therapy with recombinant GH and therapy with estrogens/progestins	1/++0
D2	BMI is not sufficiently reliable to identify TS subjects with obesity, therefore waist circumference, waist-to-height ratio and body composition assessment are recommended annually	1/++0
D3	Annual screening for liver function tests is recommended, and liver ultrasound should be performed if they are markedly and persistently altered. HRT should not be interrupted because of liver dysfunction	1/++0
D4	At least one ambulatory BP assessment per year is required and, at the first diagnosis of hypertension, secondary causes must be excluded. It is reasonable that Holter BP measurement is per- formed at least once	1/+++
D5	As antihypertensive drugs β-blockers, ARBs or ACE-I should be used in first-line treatment. In addition, losartan may be effective in reducing aortic growth velocity	2/+00
D6	During transition, TTE or cardiac magnetic resonance CMR assess- ment studies should be performed and tailored to each patient in case of structural heart disease	1/++0

magnetic resonance; *DM* diabetes mellitus; *ERT* estrogen replacement therapy; *FT4* free thyroxine; *FPG* fasting plasma glucose; *GH* growth hormone; *HbA1c* glycated hemoglobin; *HRT* hormone replacement therapy; *OGTT* oral glucose tolerance test; *QoL* quality of life; *TPO* thyroid peroxidase; *TS* Turner syndrome; *TSH* thyroid-stimulating hormone; *TTE* trans-thoracic echocardiography
[27], transdermal preparations of estradiol should be the (Statement B5). However, in prepubertal girls with TS a

[27], transdermal preparations of estradiof should be the preferred choice (Statement B2). In fact, ERT administered via a systemic route tends to mimic physiological estradiol secretion more closely. In contrast, when given orally estrogen reaches the systemic circulation only after absorption through the portal venous system, thus exposing the liver to a greater amount of estrogen than the rest of the body [27]. However, the oral route might be the preferred choice from the patients' perspective and often girls ask for an oral estroprogestin pills as maintenance HRT at the end of induction.

Girls with TS usually have a normal uterus, mandating the addition of progestin after 2 years of ERT or on the occurrence of breakthrough bleeding (Statement B3) [5]. Although a 10-day treatment cycle and natural progesterone preparations are preferred, there is a paucity of evidence to support these options [5, 28].

Fertility is an area of considerable concern for individuals with TS [5], and thus fertility counseling should be undertaken as soon as diagnosis is confirmed. At transition, fertility counseling should be repeated, or conducted if it has not been undertaken before. Those with TS should be counseled that their ability to conceive decreases with age and that spontaneous pregnancies are rare (Statement B4) [5]. A fertility assessment should be performed at transition, even if conducted previously, and periodically thereafter until premature ovarian insufficiency occurs reliable marker predicting absent pubertal ovarian activity and thus indicating the need for precocious fertility preservation is not available, even if some evidence may suggest that undetectable anti-Mullerian hormone levels predict absent pubertal onset [29]. Young women with TS and persistent ovarian function should be counseled that oocyte cryopreservation after controlled ovarian hyperstimulation or embryo cryopreservation are the best options available to preserve fertility (Statement B5) [30, 31]. Ovarian tissue cryopreservation (OTC) is an established option for those undergoing treatments that can damage the ovaries (Statement B5) and may represent the only option for prepubertal TS patients, although it should still be regarded as experimental [32]. To date, two pregnancies have been reported in patients with TS after implantation of ovarian tissue, one following autografting of ovarian tissue donated by her monozygotic twin and, more recently, a clinical pregnancy conceived naturally after re-implantation of cryopreserved ovarian tissue removed soon after spontaneous puberty [33, 34]. In addition to surgical risks and ethical issues, removal of one or part of an ovary may cause earlier premature ovarian insufficiency. All these issues, in the light of current knowledge, should be discussed with the prepubertal adolescent and her family. Since fertility preservation procedures are also invasive,

Table 1 (continued)				
No	Statement	Rating		
D7	Patients' awareness of the risk of aortic dissections and the early recognition of symptoms are essential	1/++0		
D8	Surgical management of the aortic root and ascending aorta is recommended for women with TS aged \geq 15 years, ascending ASI \geq 25 mm/m ² , regardless of the presence of risk factors for aortic dissection	2/++0		
D9	The assessment of anti-TPO antibodies is recommended annually, until positive results. The assessment of TSH \pm FT4 is recom- mended annually, regardless of the presence of Hashimoto's thyroiditis or L-thyroxine replacement therapy	1/+++		
D10	At transition, thyroid ultrasound should be performed and repeated every two years in cases of altered thyroid function or positive Anti-TPO Ab, or annually in case of thyroid nodules and/or goiters	2/++0		
D11	Screening for celiac disease should be performed once every 3 years or earlier in case of clinical suspicion. Autoimmunity screening for the other diseases should be performed only in case of clinical suspicion	1/++0		
D12	Audiologic and sensorineural screening should be periodically per- formed in all patients to monitor hearing loss and other complica- tions, and to guarantee psychosocial integration and improved QoL	1/++0		

ACE-I angiotensin-converting enzyme inhibitor; ARB angiotensin receptor blocker; ASI aortic size index; BMI body mass index; CMR cardiac

potentially fraught with complications and costly, patients with TS should also be counseled about alternative paths to parenthood.

Bone health

Osteopenia and osteoporosis affect approximately 50% of women with TS (Statement B6) [35, 36]. Patients with TS have a 25% increased fracture risk compared with the general population, predominantly seen during childhood and after 45 years [37, 38]. Dual-energy X-ray absorptiometry (DXA) is the gold standard for assessment of bone mineral density (BMD), although it has some limitations in TS patients. Areal BMD values obtained by DXA are influenced by bone size and short stature, thus resulting in an underestimation of BMD. Healthy lifestyle advice, including on physical activity and nutrition, should be given to patients with TS, and patients with TS should be screened for vitamin D deficiency between 9 and 11 years of age, and every 2–3 years thereafter [39].

ERT plays a pivotal role in preserving bone health in individuals with TS (Statement B7) [40–44]. The timing of ERT initiation is crucial in safeguarding bone health in patients with TS [40, 42–44]. It appears that the use of transdermal estrogens may have a more favorable impact on bone mass compared with oral administration (Statement B8) [45].

Oncological risk

There is a lack of studies specifically focusing on cancer risk during the transition age in TS, with existing information primarily derived from large-scale population-based studies, which encompass patients within the transition age range [46–50]. Standardized incidence ratios range from 0.9 to 1.82, with increased risk of benign central nervous system tumors and skin neoplasms, and a reduced risk for breast cancer [46–50]. HRT and growth hormone treatments do not seem to affect overall cancer risk, including breast cancer, in females with TS, however, it is important to closely monitor for specific types of tumors such as thyroid carcinoma, meningioma and skin tumors (Statement B9) [49].

International guidelines advocate for prophylactic gonadectomy in patients with TS with a Y chromosome (TS + Y)upon diagnosis [5, 51]. However, only one national registry reported an increased risk of gonadoblastoma by the age of 25 in such individuals [47], and the quality of evidence behind this recommendation is rather low. Most of the germ cell tumors in patients with TS + Y described thus far are based on a small number of case reports and case series with likely publication bias, and the optimal strategy for screening and performing gonadectomy in these patients remains unclear. Cases of gonadoblastoma have been described in children (two reports of dysgerminoma at 6 and 10 years old) but generally malignant transformation occurs in older patients in their second decade [51]. A recent multicenter study evaluating 44 patients with TS + Y (19 patients aged \geq 13 years) found that 42% of girls entered puberty spontaneously and 11% had spontaneous menarche, supporting gonadal function [52]. Gonadoblastoma was identified in seven patients (19%), one patient had in situ germ cell neoplasia and one patient had dysgerminoma (3%) [52]. In addition, a number of reports of spontaneous pregnancies and live births have been published, demonstrating that some women with TS + Y have gonadal function [53–55]. As fertility preservation methods improve, the potential for fertility in patients with TS + Y needs to be weighed against the risk of gonadal malignancy, and discussions regarding gonadectomy conducted with these patients and their caregivers (Statement B10) [53, 54].

Social and psychological aspects

Although historically physicians have often withheld information from patients with sex development disorders (including TS), the current guidelines advocate for patients' rights to be informed of their diagnosis [5, 56, 57]. Open and transparent communication with physicians is an essential part of transitioning from pediatric to adult care for patients with TS as it promotes self-awareness and identity formation [56]. The Chicago Consensus Statement underscores the importance of gradual disclosure tailored to the patient's cognitive and emotional capacities [57]. Involvement of the patient in their own treatment path is fundamental (Statement C1).

Individuals with TS typically have normal IQ levels but exhibit deficits in visual-spatial reasoning, information processing speed, working memory, executive functions and social cognition, which impact their daily life [58–60]. These deficits also significantly affect academic performance, social interactions and occupational functioning (Statement C2) [58–60]. Although HRT can improve some aspects of cognitive functioning, social competencies remain affected [58].

There is an increased prevalence of psychiatric disorders, especially anxiety and depression, in individuals with TS (Statement C3) [61–64]. Cultural factors, physiological aspects and social interactions influence emotional wellbeing in adults with TS [65]. Anxiety and depression impair the psychological health and functioning of patients with TS (Statement C4), with anxiety exacerbating attention challenges [3].

Routine neuropsychological assessments and psychodiagnostic evaluations aimed at detecting and treating critical neurocognitive issues are essential in early identification of psychological and emotional problems (Statement C5), especially in adolescents transitioning to adulthood (Statement C6) [66]. Standardized questionnaires and psychometric tools are recommended for screening symptoms of anxiety and depression, and any detected symptoms should prompt further assessment by mental health specialists [63, 67].

Addressing sexuality-related attitudes and behaviors in adolescents with TS prior to transitioning to adult care is vital for improved psychosocial well-being (Statement C7). Inducing puberty in alignment with physiological norms is crucial, and psychological support is especially important for patients from low socioeconomic backgrounds [68, 69].

Patients with TS often have lower QoL, particularly in psychosocial aspects, despite maintaining a stable socioeconomic status (Statement C8) [70, 71]. Life satisfaction is influenced by personal factors, and attention to self-perception and health-related sentiments is critical for improving QoL [72].

There is an increased prevalence of eating disorders among adolescents with TS, and aspects such as negative self-image, immature behavior, delayed pubertal development and HRT may contribute to their etiology (Statement C9) [73, 74].

Systemic and metabolic disorders

Patients with TS frequently have various metabolic, cardiovascular and autoimmune disorders.

TS is associated with an increased incidence and prevalence of diabetes mellitus (DM) [75–78]. DM in patients with TS is primarily caused by impaired insulin secretion due to autoimmunity or other reasons that are currently unknown [76, 77]. The time-course of insulin, glucose, and insulinogenic index in response to glycemic challenge do not differ between antibody-positive and antibody-negative individuals with DM [79]; therefore, screening for autoimmune diabetes is not recommended until DM is diagnosed. The available evidence indicates that the oral glucose tolerance test (OGTT) is superior to fasting glucose and glycated hemoglobin (HbA1c) in early DM diagnosis [77, 78, 80]. Therefore, an annual HbA1c and fasting glucose screening and a biennial OGTT screening are recommended (Statement D1).

Obesity is common in patients with TS, who have a distinct body fat distribution, characterized by increased visceral fat and reduced subcutaneous fat and lean body mass compared with women without TS [81–86]. Therefore, body mass index (BMI) is not an effective measure of adiposity in this population and alternative measures, such as waist circumference, waist-to-height ratio and body composition analysis, should be used (Statement D2) [86].

Hepatic abnormalities, including architectural changes, alterations in liver enzymes and hepatic steatosis, are common in TS [87–91]. Their cause remains uncertain, but

congenital vascular disorders, metabolic syndrome and estrogen deficiency, environmental influences, lifestyle, genetic and epigenetic factors, may play a role [89, 91]. In addition, biliary conditions, including sclerosing cholangitis, primary biliary cholangitis, biliary atresia and a paucity of bile ducts, can develop, primarily as a result of primary hepatic vascular abnormalities [89, 92]. Liver function should be assessed annually in patients with TS and an ultrasound should be performed in case of significant and consistent abnormalities (Statement D3). Discontinuation of HRT is unwarranted as it does not cause deterioration of liver function [89, 91].

Patients with TS often have hypertension, affecting up to 25% of adolescents and between 13 and 58% of adults, and influenced greatly by factors such as race and lifestyle [93–97]. Hypertension in TS is predominantly characterized by an elevated systolic pressure and the absence of the typical circadian variation, with abnormal nocturnal dips observed as early as infancy [93, 97]. Patients with TS should undergo at least one ambulatory blood pressure assessment annually (Statement D4). Upon an initial diagnosis of hypertension, secondary causes should be meticulously investigated and ruled out. Beta-blockers, angiotensin receptor blockers (ARBs) or angiotensin-converting enzyme inhibitors (ACE-I) are recommended as first-line treatment for hypertension in patients with TS, while losartan can be used to reduce the growth rate of the aorta (Statement D5).

Approximately, 30% of patients with TS have congenital cardiac anomalies, with the most common being a bicuspid aortic valve and coarctation of the aorta [94, 98–100]. These anomalies are observed more frequently in those with 45X monosomy compared with those with mosaicism and often contribute to early mortality even after surgical or percutaneous intervention [98, 99, 101, 102]. In patients with TS, electrocardiogram (ECG) abnormalities such as QT interval prolongation have been reported starting from childhood, although they are generally mild in severity [94, 103].

Aortic dilation in TS is serious and often leads to fatal dissections [94, 104–106]. Typically originating in the ascending aorta, aortic dissections account for 2-8% of deaths in patients with TS [94, 107]. The condition occurs at an average age of 35 in those with TS, as opposed to 68 in the general population [94, 104, 108]. Aortic size index (ASI) is the primary parameter for assessing the risk of dissection [109]. During the transition phase, it is imperative for patients with TS to undergo trans-thoracic echocardiography (TTE) or cardiac magnetic resonance (CMR) assessments, as these methods provide high accuracy in measuring aortic size (Statement D6) [100, 110]. Educating affected patients about the risks of aortic dissections and the importance of early symptom recognition is essential (Statement D7). For those with TS aged \geq 15 years, it is advisable to undertake surgical intervention for the aortic root and ascending aorta if ascending ASI is ≥ 25 mm/m², irrespective of the presence of risk factors for aortic dissection (Statement D8).

Autoimmune disorders are also prevalent in TS due to complex genetic and environmental interactions [111–113]. Hashimoto's thyroiditis is common, affecting approximately 50% of individuals with TS [111–113]. Moreover, a Danish study revealed that 57% women with TS had at least one type of autoantibody, most commonly anti-thyroid peroxidase antibodies (TPO), which is significantly higher than in the general population [111]. The annual evaluation of anti-TPO antibodies is advisable for all patients with TS until positive results are obtained. Thyroid-stimulating hormone (TSH) and possibly free thyroxine (FT4) should be annually measured regardless of the presence of Hashimoto's thyroiditis (Statement D9). In the presence of thyroid nodules and/or goiter, an ultrasound should be conducted annually (Statement D10). Screening for celiac disease should be performed every 3 years, while screening for other autoimmune diseases should be conducted in the presence of clinical indications (Statement D11).

Furthermore, it is imperative to conduct regular audiologic and sensorineural assessments for all patients to keep track of hearing loss and related complications. These assessments are crucial for ensuring psychosocial integration and enhancing QoL (Statement D12).

Reassuringly, HRT does not appear to adversely affect the risk of DM, liver enzyme abnormalities or autoimmune conditions [114–116].

Conclusions

The TRAMITI project produced a comprehensive set of consensus statements crafted through collaborative efforts of expert Italian physicians specializing in fields related to the management of TS. These statements address four key domains: transition and referral; sexual, bone health and oncological risks; social and psychological aspects; and systemic and metabolic disorders. These consensus statements have the potential to enhance TS management by providing healthcare practitioners with a structured guide that emphasizes patient-centered care, early intervention and long-term monitoring. In addition, by focusing on the needs and psychological well-being of patients with TS, these recommendations address not only the medical, but also the psychosocial aspects that are crucial for improving the QoL and long-term outcomes of these individuals.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s40618-024-02315-4.

Acknowledgements We would like to thank Georgii Filatov and Sheridan Henness, who provided medical writing assistance prior to and after submission, respectively, on behalf of Springer Healthcare Communications. This medical writing assistance was funded by Sandoz.

Author contributions TA and MW: conception and design of the study. All the authors: members of the working group, prepared and revised the statements, analyzed and interpreted the results, and critically reviewed and approved the final manuscript for submission. All the listed authors have made substantial contributions to the conception or design of the work; or the acquisition, analysis, or interpretation of data; drafted the work or revised it critically for important intellectual content; approved the version to be published; and agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Funding This study and development of the manuscript were supported by Sandoz.

Data availability Data sharing is not applicable to this article as no datasets were generated or analyzed during the current study.

Declarations

Conflict of interest Tommaso Aversa has consulted for Pfizer and Sandoz. Anna Balducci, Roberta D'Aprile, Carolina Di Somma, Maria Felicia Faienza, Alessandra Gambineri, Claudia Giavoli, Armando Grossi, Maria Cristina Meriggiola, Eriselda Profka, Mariacarolina Salerno, and Maria Chiara Zatelli have consulted for Sandoz. Stefano Stagi has consulted for Aboca, Novonordisk, and Sandoz. Malgorzata Wasniewska is a consultant for Merck, Novonordisk, Pfizer, and Sandoz. Emanuela Scarano has consulted for Alexion, Biomarin, Novonordisk, Pfizer, and Sandoz. Luisa De Sanctis has no interests to declare.

Consent to participate Not applicable.

Consent to publication Not applicable.

Human or animal rights The study does not involve human participants and/or animals.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Ranke MB, Saenger P (2001) Turner's syndrome. Lancet 358(9278):309–314. https://doi.org/10.1016/s0140-6736(01) 05487-3
- Sybert VP, McCauley E (2004) Turner's syndrome. N Engl J Med 351(12):1227–1238. https://doi.org/10.1056/NEJMra030360
- 3. Hutaff-Lee C, Bennett E, Howell S, Tartaglia N (2019) Clinical developmental, neuropsychological, and social-emotional

features of Turner syndrome. Am J Med Genet C Semin Med Genet 181(1):126–134. https://doi.org/10.1002/ajmg.c.31687

- Gravholt CH, Viuff MH, Brun S, Stochholm K, Andersen NH (2019) Turner syndrome: mechanisms and management. Nat Rev Endocrinol 15(10):601–614. https://doi.org/10.1038/ s41574-019-0224-4
- Gravholt CH, Andersen NH, Conway GS, Dekkers OM, Geffner ME, Klein KO, Lin AE, Mauras N, Quigley CA, Rubin K, Sandberg DE, Sas TCJ, Silberbach M, Söderström-Anttila V, Stochholm K, van Alfen-van derVelden JA, Woelfle J, Backeljauw PF (2017) Clinical practice guidelines for the care of girls and women with Turner syndrome: proceedings from the 2016 Cincinnati International Turner Syndrome Meeting. Eur J Endocrinol 177(3):G1–G70. https://doi.org/10.1530/eje-17-0430
- Gawlik A, Malecka-Tendera E (2014) Transitions in endocrinology: treatment of Turner's syndrome during transition. Eur J Endocrinol 170(2):R57-74. https://doi.org/10.1530/ eje-13-0900
- Bernard V, Donadille B, Le Poulennec T, Nedelcu M, Martinerie L, Christin-Maitre S (2019) Management of endocrine disease: transition of care for young adult patients with Turner syndrome. Eur J Endocrinol 180(1):R1–R7. https://doi.org/10. 1530/eje-18-0238
- Devernay M, Ecosse E, Coste J, Carel JC (2009) Determinants of medical care for young women with Turner syndrome. J Clin Endocrinol Metab 94(9):3408–3413. https://doi.org/10.1210/jc. 2009-0495
- Ertl DA, Gleiss A, Schubert K, Culen C, Hauck P, Ott J, Gessl A, Haeusler G (2018) Health status, quality of life and medical care in adult women with Turner syndrome. Endocr Connect 7(4):534–543. https://doi.org/10.1530/ec-18-0053
- Swiglo BA, Murad MH, Schünemann HJ, Kunz R, Vigersky RA, Guyatt GH, Montori VM (2008) A case for clarity, consistency, and helpfulness: state-of-the-art clinical practice guidelines in endocrinology using the grading of recommendations, assessment, development, and evaluation system. J Clin Endocrinol Metab 93(3):666–673. https://doi.org/10.1210/jc.2007-1907
- 11. van Trotsenburg P, Stoupa A, Léger J, Rohrer T, Peters C, Fugazzola L, Cassio A, Heinrichs C, Beauloye V, Pohlenz J, Rodien P, Coutant R, Szinnai G, Murray P, Bartés B, Luton D, Salerno M, de Sanctis L, Vigone M, Krude H, Persani L, Polak M (2021) Congenital hypothyroidism: a 2020–2021 consensus guidelines update-an ENDO-European reference network initiative endorsed by the European society for pediatric endocrinology and the European society for endocrinology. Thyroid 31(3):387–419. https://doi.org/10.1089/thy.2020.0333
- Culen C, Herle M, Ertl DA, Fröhlich-Reiterer E, Blümel P, Wagner G, Häusler G (2020) Less ready for adulthood? Turner syndrome has an impact on transition readiness. Clin Endocrinol (Oxf) 93(4):449–455. https://doi.org/10.1111/cen.14293
- Streur CS, Floody EA, Lapham ZK, Sandberg DE (2022) The transition to independence and adult care for women with Turner syndrome: current status and priorities of 1338 women and parents. Am J Med Genet A 188(2):400–413. https://doi.org/10. 1002/ajmg.a.62564
- Kosteria I, Kanaka-Gantenbein C (2018) Turner syndrome: transition from childhood to adolescence. Metabolism 86:145–153. https://doi.org/10.1016/j.metabol.2017.12.016
- Endocrine Society (2023) Turner syndrome. https://www. endocrine.org/improving-practice/transitions/turner-syndrome. Accessed 12 June 2023
- Sheanon NM, Beal SJ, Kichler JC, Casnellie L, Backeljauw P, Corathers S (2020) Readiness for transition to adult care in adolescents and young adults with Turner syndrome. J Pediatr Endocrinol Metab 33(9):1165–1171. https://doi.org/10.1515/ jpem-2020-0155

- Lucaccioni L, Wong SC, Smyth A, Lyall H, Dominiczak A, Ahmed SF, Mason A (2015) Turner syndrome–issues to consider for transition to adulthood. Br Med Bull 113(1):45–58. https:// doi.org/10.1093/bmb/ldu038
- Sakakibara H (2017) Transition of women with Turner syndrome from pediatrics to adult health care: current situation and associated problems. Front Pediatr 5:28. https://doi.org/10.3389/fped. 2017.00028
- Shankar RK, Backeljauw PF (2018) Current best practice in the management of Turner syndrome. Ther Adv Endocrinol Metab 9(1):33–40. https://doi.org/10.1177/2042018817746291
- Gravholt CH (2019) New international guidelines on Turner syndrome. Horm Res Paediatr 91(Suppl. 1):14. https://doi.org/10. 1159/000501868
- Sharma N, O'Hare K, Antonelli RC, Sawicki GS (2014) Transition care: future directions in education, health policy, and outcomes research. Acad Pediatr 14(2):120–127. https://doi.org/10. 1016/j.acap.2013.11.007
- 22. Lin AE, Prakash SK, Andersen NH, Viuff MH, Levitsky LL, Rivera-Davila M, Crenshaw ML, Hansen L, Colvin MK, Hayes FJ, Lilly E, Snyder EA, Nader-Eftekhari S, Aldrich MB, Bhatt AB, Prager LM, Arenivas A, Skakkebaek A, Steeves MA, Kreher JB, Gravholt CH (2019) Recognition and management of adults with Turner syndrome: from the transition of adolescence through the senior years. Am J Med Genet A 179(10):1987– 2033. https://doi.org/10.1002/ajmg.a.61310
- Zahra B, Lyall H, Sastry A, Freel EM, Dominiczak AF, Mason A (2021) Evaluating transition in Turner syndrome in the West of Scotland. J Pediatr Endocrinol Metab 34(4):473–477. https:// doi.org/10.1515/jpem-2020-0242
- 24. Davidse K, van Staa A, Geilvoet W, van Eck JP, Pellikaan K, Baan J, Hokken-Koelega ACS, van den Akker ELT, Sas T, Hannema SE, van der Lely AJ, de Graaff LCG (2022) We mind your step: understanding and preventing drop-out in the transfer from paediatric to adult tertiary endocrine healthcare. Endocr Connect 11(5):e220025. https://doi.org/10.1530/ec-22-0025
- Rubin KR (2008) Turner syndrome: transition from pediatrics to adulthood. Endocr Pract 14(6):775–781. https://doi.org/10.4158/ ep.14.6.775
- Got Transition (2020) Six core elements of health care transition 3.0. https://www.gottransition.org/six-core-elements/. Accessed 12 June 2023
- Bondy CA (2007) Care of girls and women with Turner syndrome: a guideline of the Turner syndrome study group. J Clin Endocrinol Metab 92(1):10–25. https://doi.org/10.1210/jc. 2006-1374
- 28. Nordenström A, Ahmed SF, van den Akker E, Blair J, Bonomi M, Brachet C, Broersen LHA, Claahsen-van der Grinten HL, Dessens AB, Gawlik A, Gravholt CH, Juul A, Krausz C, Raivio T, Smyth A, Touraine P, Vitali D, Dekkers OM (2022) Pubertal induction and transition to adult sex hormone replacement in patients with congenital pituitary or gonadal reproductive hormone deficiency: an Endo-ERN clinical practice guideline. Eur J Endocrinol 186(6):G9–G49. https://doi.org/10.1530/eje-22-0073
- Lunding SA, Aksglaede L, Anderson RA, Main KM, Juul A, Hagen CP, Pedersen AT (2015) AMH as predictor of premature ovarian insufficiency: a longitudinal study of 120 Turner syndrome patients. J Clin Endocrinol Metab 100(7):E1030-1038. https://doi.org/10.1210/jc.2015-1621
- Hagen CP, Mouritsen A, Mieritz MG, Tinggaard J, Wohlfart-Veje C, Fallentin E, Brocks V, Sundberg K, Jensen LN, Anderson RA, Juul A, Main KM (2015) Circulating AMH reflects ovarian morphology by magnetic resonance imaging and 3D ultrasound in 121 healthy girls. J Clin Endocrinol Metab 100(3):880–890. https://doi.org/10.1210/jc.2014-3336

- Vergier J, Bottin P, Saias J, Reynaud R, Guillemain C, Courbiere B (2019) Fertility preservation in Turner syndrome: karyotype does not predict ovarian response to stimulation. Clin Endocrinol (Oxf) 91(5):646–651. https://doi.org/10.1111/cen.14076
- 32. Schleedoorn MJ, Mulder BH, Braat DDM, Beerendonk CCM, Peek R, Nelen W, Van Leeuwen E, Van der Velden A, Fleischer K, Turner Fertility Expert Panel O (2020) International consensus: ovarian tissue cryopreservation in young Turner syndrome patients: outcomes of an ethical Delphi study including 55 experts from 16 different countries. Hum Reprod 35(5):1061– 1072. https://doi.org/10.1093/humrep/deaa007
- Dunlop CE, Jack SA, Telfer EE, Zahra S, Anderson RA (2023) Clinical pregnancy in Turner syndrome following re-implantation of cryopreserved ovarian cortex. J Assist Reprod Genet 40(10):2385–2390. https://doi.org/10.1007/s10815-023-02905-w
- 34. Donnez J, Dolmans MM, Squifflet J, Kerbrat G, Jadoul P (2011) Live birth after allografting of ovarian cortex between monozygotic twins with Turner syndrome (45, XO/46, XX mosaicism) and discordant ovarian function. Fertil Steril 96(6):1407–1411. https://doi.org/10.1016/j.fertnstert.2011.09.012
- Benetti-Pinto CL, Bedone A, Magna LA, Marques-Neto JF (2002) Factors associated with the reduction of bone density in patients with gonadal dysgenesis. Fertil Steril 77(3):571–575. https://doi.org/10.1016/s0015-0282(01)03215-0
- Gravholt CH, Lauridsen AL, Brixen K, Mosekilde L, Heickendorff L, Christiansen JS (2002) Marked disproportionality in bone size and mineral, and distinct abnormalities in bone markers and calcitropic hormones in adult turner syndrome: a cross-sectional study. J Clin Endocrinol Metab 87(6):2798–2808. https://doi.org/ 10.1210/jcem.87.6.8598
- Wasserman H, Backeljauw PF, Khoury JC, Kalkwarf HJ, Gordon CM (2018) Bone fragility in Turner syndrome: fracture prevalence and risk factors determined by a national patient survey. Clin Endocrinol (Oxf) 89(1):46–55. https://doi.org/10.1111/cen. 13614
- Han TS, Cadge B, Conway GS (2006) Hearing impairment and low bone mineral density increase the risk of bone fractures in women with Turner's syndrome. Clin Endocrinol (Oxf) 65(5):643–647. https://doi.org/10.1111/j.1365-2265.2006. 02643.x
- Chiarito M, Brunetti G, D'Amato G, Faienza MF (2020) Monitoring and maintaining bone health in patients with Turner syndrome. Expert Rev Endocrinol Metab 15(6):431–438. https://doi. org/10.1080/17446651.2020.1834846
- 40. Nakamura T, Tsuburai T, Tokinaga A, Nakajima I, Kitayama R, Imai Y, Nagata T, Yoshida H, Hirahara F, Sakakibara H (2015) Efficacy of estrogen replacement therapy (ERT) on uterine growth and acquisition of bone mass in patients with Turner syndrome. Endocr J 62(11):965–970. https://doi.org/10.1507/ endocrj.EJ15-0172
- Cleemann L, Holm K, Kobbernagel H, Kristensen B, Skouby SO, Jensen AK, Gravholt CH (2017) Dosage of estradiol, bone and body composition in Turner syndrome: a 5-year randomized controlled clinical trial. Eur J Endocrinol 176(2):233–242. https:// doi.org/10.1530/eje-16-0582
- Nguyen HH, Wong P, Strauss BJ, Jones G, Ebeling PR, Milat F, Vincent A (2017) Delay in estrogen commencement is associated with lower bone mineral density in Turner syndrome. Climacteric 20(5):436–441. https://doi.org/10.1080/13697137.2017.1325461
- 43. Cameron-Pimblett A, Davies MC, Burt E, Talaulikar VS, La Rosa C, King TFJ, Conway GS (2019) Effects of estrogen therapies on outcomes in Turner syndrome: assessment of induction of puberty and adult estrogen use. J Clin Endocrinol Metab 104(7):2820–2826. https://doi.org/10.1210/jc.2018-02137
- 44. Nishigaki S, Itonaga T, Hasegawa Y, Kawai M (2021) Starting age of oestrogen-progestin therapy is negatively associated with

bone mineral density in young adults with Turner syndrome independent of age and body mass index. Clin Endocrinol (Oxf) 95(1):84–91. https://doi.org/10.1111/cen.14484

- 45. Zaiem F, Alahdab F, Al Nofal A, Murad MH, Javed A (2017) Oral versus transdermal estrogen in Turner syndrome: a systematic review and meta-analysis. Endocr Pract 23(4):408–421. https://doi.org/10.4158/ep161622.Or
- Hasle H, Olsen JH, Nielsen J, Hansen J, Friedrich U, Tommerup N (1996) Occurrence of cancer in women with Turner syndrome. Br J Cancer 73(9):1156–1159. https://doi.org/10.1038/bjc.1996. 222
- Schoemaker MJ, Swerdlow AJ, Higgins CD, Wright AF, Jacobs PA (2008) Cancer incidence in women with Turner syndrome in Great Britain: a national cohort study. Lancet Oncol 9(3):239– 246. https://doi.org/10.1016/s1470-2045(08)70033-0
- 48. Ji J, Zöller B, Sundquist J, Sundquist K (2016) Risk of solid tumors and hematological malignancy in persons with Turner and Klinefelter syndromes: a national cohort study. Int J Cancer 139(4):754–758. https://doi.org/10.1002/ijc.30126
- 49. Gambineri A, Scarano E, Rucci P, Perri A, Tamburrino F, Altieri P, Corzani F, Cecchetti C, Dionese P, Belardinelli E, Ibarra-Gasparini D, Menabò S, Vicennati V, Repaci A, di Dalmazi G, Pelusi C, Zavatta G, Virdi A, Neri I, Fanelli F, Mazzanti L, Pagotto U (2022) New insights into the comorbid conditions of Turner syndrome: results from a long-term monocentric cohort study. J Endocrinol Invest 45(12):2247– 2256. https://doi.org/10.1007/s40618-022-01853-z
- 50. Kim SE, Park SH, Han K, Cho WK, Suh BK, Park YG (2022) Population prevalence, cancer risk, and mortality risk of Turner syndrome in South Korean women based on national health insurance service data. Yonsei Med J 63(11):991–998. https://doi.org/10.3349/ymj.2022.0143
- Pyle LC, Nathanson KL (2017) A practical guide for evaluating gonadal germ cell tumor predisposition in differences of sex development. Am J Med Genet C Semin Med Genet 175(2):304–314. https://doi.org/10.1002/ajmg.c.31562
- Dabrowski E, Johnson EK, Patel V, Hsu Y, Davis S, Goetsch AL, Habiby R, Brickman WJ, Finlayson C (2020) Turner syndrome with Y chromosome: spontaneous thelarche, menarche, and risk of malignancy. J Pediatr Adolesc Gynecol 33(1):10– 14. https://doi.org/10.1016/j.jpag.2019.08.011
- 53. Dumic M, Lin-Su K, Leibel NI, Ciglar S, Vinci G, Lasan R, Nimkarn S, Wilson JD, McElreavey K, New MI (2008) Report of fertility in a woman with a predominantly 46, XY karyotype in a family with multiple disorders of sexual development. J Clin Endocrinol Metab 93(1):182–189. https://doi.org/10. 1210/jc.2007-2155
- Landin-Wilhelmsen K, Bryman I, Hanson C, Hanson L (2004) Spontaneous pregnancies in a Turner syndrome woman with Y-chromosome mosaicism. J Assist Reprod Genet 21(6):229– 230. https://doi.org/10.1023/b:jarg.0000040239.40913.c3
- 55. Portnoi MF, Chantot-Bastaraud S, Christin-Maitre S, Carbonne B, Beaujard MP, Keren B, Levy J, Dommergues M, Cabrol S, Hyon C, Siffroi JP (2012) Familial Turner syndrome with an X;Y translocation mosaicism: implications for genetic counseling. Eur J Med Genet 55(11):635–640. https://doi.org/10. 1016/j.ejmg.2012.07.001
- D'Alberton F (2010) Disclosing disorders of sex development and opening the doors. Sex Dev 4(4–5):304–309. https://doi. org/10.1159/000317480
- Hughes IA, Houk C, Ahmed SF, Lee PA (2006) Consensus statement on management of intersex disorders. Arch Dis Child 91(7):554–563. https://doi.org/10.1136/adc.2006.098319
- 58. Hong DS, Dunkin B, Reiss AL (2011) Psychosocial functioning and social cognitive processing in girls with Turner

syndrome. J Dev Behav Pediatr 32(7):512–520. https://doi.org/ 10.1097/DBP.0b013e3182255301

- Lepage JF, Dunkin B, Hong DS, Reiss AL (2011) Contribution of executive functions to visuospatial difficulties in prepubertal girls with Turner syndrome. Dev Neuropsychol 36(8):988– 1002. https://doi.org/10.1080/87565641.2011.584356
- Hong DS, Reiss AL (2012) Cognition and behavior in Turner syndrome: a brief review. Pediatr Endocrinol Rev 9(Suppl 2):710–712
- Lesniak-Karpiak K, Mazzocco MM, Ross JL (2003) Behavioral assessment of social anxiety in females with Turner or fragile X syndrome. J Autism Dev Disord 33(1):55–67. https://doi. org/10.1023/a:1022230504787
- Cardoso G, Daly R, Haq NA, Hanton L, Rubinow DR, Bondy CA, Schmidt P (2004) Current and lifetime psychiatric illness in women with Turner syndrome. Gynecol Endocrinol 19(6):313–319. https://doi.org/10.1080/09513590400021227
- Kiliç BG, Ergür AT, Ocal G (2005) Depression, levels of anxiety and self-concept in girls with Turner's syndrome. J Pediatr Endocrinol Metab 18(11):1111–1117. https://doi.org/10.1515/jpem.2005.18.11.1111
- Green T, Flash S, Reiss AL (2019) Sex differences in psychiatric disorders: what we can learn from sex chromosome aneuploidies. Neuropsychopharmacology 44(1):9–21. https://doi.org/10.1038/ s41386-018-0153-2
- 65. Lašaitė L, Krikščiūnienė R, Žilaitienė B, Verkauskienė R (2019) Emotional state, cognitive functioning and quality of life of adult women with Turner syndrome in Lithuania. Growth Horm IGF Res 45:37–42. https://doi.org/10.1016/j.ghir.2019.03.001
- 66. Alexandrou E, Corathers S, Gutmark-Little I, Casnellie L, Gerstle M, Tatum J, Khoury J, Backeljauw P (2022) Improving anxiety screening in patients with Turner syndrome. Horm Res Paediatr 95(1):68–75. https://doi.org/10.1159/000524169
- Ye M, Lin H, Lash GE, Yuan L, Li L (2021) Effects of psychosomatic mutual aid treatment on anxiety and depression in Turner syndrome. Front Psychiatry 12:644147. https://doi.org/10.3389/ fpsyt.2021.644147
- Carel JC, Elie C, Ecosse E, Tauber M, Léger J, Cabrol S, Nicolino M, Brauner R, Chaussain JL, Coste J (2006) Self-esteem and social adjustment in young women with Turner syndrome–influence of pubertal management and sexuality: population-based cohort study. J Clin Endocrinol Metab 91(8):2972–2979. https:// doi.org/10.1210/jc.2005-2652
- Dołęga Z, Jeż W, Irzyniec T (2014) The cohort effect in studies related to differences in psychosocial functioning of women with Turner syndrome. Endokrynol Pol 65(4):287–294. https://doi. org/10.5603/ep.2014.0039
- Gould HN, Bakalov VK, Tankersley C, Bondy CA (2013) High levels of education and employment among women with Turner syndrome. J Womens Health (Larchmt) 22(3):230–235. https:// doi.org/10.1089/jwh.2012.3931
- Noordman ID, van der Velden JA, Timmers HJ, Reisch N, Richter-Unruh A, Pienkowksi C, Roeleveld N, Claahsen-van der Grinten HL (2021) Socioeconomic status in patients with Turner syndrome. Compr Psychoneuroendocrinol 5:100030. https://doi. org/10.1016/j.cpnec.2021.100030
- Jeż W, Tobiasz-Adamczyk B, Brzyski P, Majkowicz M, Pankiewicz P, Irzyniec TJ (2018) Social and medical determinants of quality of life and life satisfaction in women with Turner syndrome. Adv Clin Exp Med 27(2):229–236. https://doi.org/10.17219/acem/66986
- 73. Brinch M, Manthorpe T (1987) Short stature as a possible etiological factor in anorexia nervosa. Acta Psychiatr Scand 76(3):328–332. https://doi.org/10.1111/j.1600-0447.1987.tb029 03.x

- Muhs A, Lieberz K (1993) Anorexia nervosa and Turner's syndrome. Psychopathology 26(1):29–40. https://doi.org/10.1159/ 000284797
- 75. Cicognani A, Mazzanti L, Tassinari D, Pellacani A, Forabosco A, Landi L, Pifferi C, Cacciari E (1988) Differences in carbohydrate tolerance in Turner syndrome depending on age and karyotype. Eur J Pediatr 148(1):64–68. https://doi.org/10.1007/bf00441818
- Bakalov VK, Cheng C, Zhou J, Bondy CA (2009) X-chromosome gene dosage and the risk of diabetes in Turner syndrome. J Clin Endocrinol Metab 94(9):3289–3296. https://doi.org/10.1210/jc. 2009-0384
- 77. Ibarra-Gasparini D, Altieri P, Scarano E, Perri A, Morselli-Labate AM, Pagotto U, Mazzanti L, Pasquali R, Gambineri A (2018) New insights on diabetes in Turner syndrome: results from an observational study in adulthood. Endocrine 59(3):651– 660. https://doi.org/10.1007/s12020-017-1336-z
- Sheanon N, Elder D, Khoury J, Casnellie L, Gutmark-Little I, Cernich J, Backeljauw PF (2021) Increased prevalence of betacell dysfunction despite normal HbA1c in youth and young adults with Turner syndrome. Horm Res Paediatr 94(7–8):297– 306. https://doi.org/10.1159/000520233
- 79. Mitsch C, Alexandrou E, Norris AW, Pinnaro CT (2023) Hyperglycemia in Turner syndrome: Impact, mechanisms, and areas for future research. Front Endocrinol (Lausanne) 14:1116889. https://doi.org/10.3389/fendo.2023.1116889
- Choi IK, Kim DH, Kim HS (2005) The abnormalities of carbohydrate metabolism in Turner syndrome: analysis of risk factors associated with impaired glucose tolerance. Eur J Pediatr 164(7):442–447. https://doi.org/10.1007/s00431-005-1643-x
- Hanaki K, Ohzeki T, Ishitani N, Motozumi H, Matsuda-Ohtahara H, Shiraki K (1992) Fat distribution in overweight patients with Ullrich-Turner syndrome. Am J Med Genet 42(4):428–430. https://doi.org/10.1002/ajmg.1320420405
- Ohzeki T, Hanaki K, Motozumi H, Ohtahara H, Urashima H, Shiraki K (1993) Assessment of subcutaneous fat using ultrasonography in the Ullrich-Turner syndrome. Am J Med Genet 46(4):450–452. https://doi.org/10.1002/ajmg.1320460421
- Gravholt CH, Weis Naeraa R (1997) Reference values for body proportions and body composition in adult women with Ullrich-Turner syndrome. Am J Med Genet 72(4):403–408. https://doi.org/10.1002/(sici)1096-8628(19971112)72:4% 3c403::aid-ajmg6%3e3.0.co;2-r
- Ostberg JE, Thomas EL, Hamilton G, Attar MJ, Bell JD, Conway GS (2005) Excess visceral and hepatic adipose tissue in Turner syndrome determined by magnetic resonance imaging: estrogen deficiency associated with hepatic adipose content. J Clin Endocrinol Metab 90(5):2631–2635. https://doi.org/10. 1210/jc.2004-1939
- Gravholt CH, Hjerrild BE, Mosekilde L, Hansen TK, Rasmussen LM, Frystyk J, Flyvbjerg A, Christiansen JS (2006) Body composition is distinctly altered in Turner syndrome: relations to glucose metabolism, circulating adipokines, and endothelial adhesion molecules. Eur J Endocrinol 155(4):583–592. https://doi.org/10.1530/eje.1.02267
- 86. Álvarez-Nava F, Racines M, Witt J, Guarderas J, Estévez M, Lanes R (2019) Anthropometric variables as cardiovascular risk predictors in a cohort of adult subjects with Turner syndrome. Diabetes Metab Syndr Obes 12:1795–1809. https://doi. org/10.2147/dmso.S214787
- Roulot D, Valla D (2006) Hepatic disease in Turner syndrome. Int Congr Ser 1298:146–151. https://doi.org/10.1016/j.ics. 2006.06.006
- El-Mansoury M, Berntorp K, Bryman I, Hanson C, Innala E, Karlsson A, Landin-Wilhelmsen K (2008) Elevated liver enzymes in Turner syndrome during a 5-year follow-up study.

Clin Endocrinol (Oxf) 68(3):485–490. https://doi.org/10. 1111/j.1365-2265.2007.03166.x

- Roulot D (2013) Liver involvement in Turner syndrome. Liver Int 33(1):24–30. https://doi.org/10.1111/liv.12007
- Calanchini M, Moolla A, Tomlinson JW, Cobbold JF, Grossman A, Fabbri A, Turner HE (2018) Liver biochemical abnormalities in Turner syndrome: a comprehensive characterization of an adult population. Clin Endocrinol (Oxf) 89(5):667–676. https://doi.org/10.1111/cen.13811
- Roulot D, Degott C, Chazouillères O, Oberti F, Calès P, Carbonell N, Benferhat S, Bresson-Hadni S, Valla D (2004) Vascular involvement of the liver in Turner's syndrome. Hepatology 39(1):239–247. https://doi.org/10.1002/hep.20026
- Fedor I, Zold E, Barta Z (2022) Liver abnormalities in Turner syndrome: the importance of estrogen replacement. J Endocr Soc 6(10):bvac124. https://doi.org/10.1210/jendso/bvac124
- 93. Nathwani NC, Unwin R, Brook CG, Hindmarsh PC (2000) Blood pressure and Turner syndrome. Clin Endocrinol (Oxf) 52(3):363–370. https://doi.org/10.1046/j.1365-2265.2000. 00960.x
- Dulac Y, Pienkowski C, Abadir S, Tauber M, Acar P (2008) Cardiovascular abnormalities in Turner's syndrome: what prevention? Arch Cardiovasc Dis 101(7–8):485–490. https://doi.org/ 10.1016/j.acvd.2008.05.007
- Turtle EJ, Sule AA, Bath LE, Denvir M, Gebbie A, Mirsadraee S, Webb DJ (2013) Assessing and addressing cardiovascular risk in adults with Turner syndrome. Clin Endocrinol (Oxf) 78(5):639– 645. https://doi.org/10.1111/cen.12104
- 96. De Groote K, Devos D, Van Herck K, Demulier L, Buysse W, De Schepper J, De Wolf D (2015) Abnormal aortic arch morphology in Turner syndrome patients is a risk factor for hypertension. Heart Vessels 30(5):618–625. https://doi.org/10.1007/ s00380-014-0529-0
- 97. De Groote K, Demulier L, De Backer J, De Wolf D, De Schepper J, T'sjoen G, De Backer T (2015) Arterial hypertension in Turner syndrome: a review of the literature and a practical approach for diagnosis and treatment. J Hypertens 33(7):1342–1351. https://doi.org/10.1097/hjh.00000000000599
- Mazzanti L, Cacciari E (1998) Congenital heart disease in patients with Turner's syndrome. Italian study group for Turner syndrome (ISGTS). J Pediatr 133(5):688–692. https://doi.org/10. 1016/s0022-3476(98)70119-2
- Völkl TM, Degenhardt K, Koch A, Simm D, Dörr HG, Singer H (2005) Cardiovascular anomalies in children and young adults with Ullrich-Turner syndrome the Erlangen experience. Clin Cardiol 28(2):88–92. https://doi.org/10.1002/clc.4960280209
- 100. Silberbach M, Roos-Hesselink JW, Andersen NH, Braverman AC, Brown N, Collins RT, De Backer J, Eagle KA, Hiratzka LF, Johnson WH Jr, Kadian-Dodov D, Lopez L, Mortensen KH, Prakash SK, Ratchford EV, Saidi A, van Hagen I, Young LT (2018) Cardiovascular health in Turner syndrome: a scientific statement from the American heart association. Circ Genom Precis Med 11(10):e000048. https://doi.org/10.1161/hcg.00000 00000000048
- 101. van den Hoven AT, Duijnhouwer AL, Eicken A, Aboulhosn J, de Bruin C, Backeljauw PF, Demulier L, Chessa M, Uebing A, Veldtman GR, Armstrong AK, van den Bosch AE, Witsenburg M, Roos-Hesselink JW (2017) Adverse outcome of coarctation stenting in patients with Turner syndrome. Catheter Cardiovasc Interv 89(2):280–287. https://doi.org/10.1002/ccd.26728
- 102. Chew JD, Hill KD, Jacobs ML, Jacobs JP, Killen SAS, Godown J, Wallace AS, Thibault D, Chiswell K, Bichell DP, Soslow JH (2019) Congenital heart surgery outcomes in Turner syndrome: the society of thoracic surgeons database analysis. Ann Thorac Surg 108(5):1430–1437. https://doi.org/10.1016/j.athoracsur. 2019.05.047

- Bondy CA, Ceniceros I, Van PL, Bakalov VK, Rosing DR (2006) Prolonged rate-corrected QT interval and other electrocardiogram abnormalities in girls with Turner syndrome. Pediatrics 118(4):e1220-1225. https://doi.org/10.1542/peds.2006-0776
- 104. Gravholt CH, Landin-Wilhelmsen K, Stochholm K, Hjerrild BE, Ledet T, Djurhuus CB, Sylvén L, Baandrup U, Kristensen B, Christiansen JS (2006) Clinical and epidemiological description of aortic dissection in Turner's syndrome. Cardiol Young 16(5):430–436. https://doi.org/10.1017/s1047951106000928
- 105. Carlson M, Airhart N, Lopez L, Silberbach M (2012) Moderate aortic enlargement and bicuspid aortic valve are associated with aortic dissection in Turner syndrome: report of the international turner syndrome aortic dissection registry. Circulation 126(18):2220–2226. https://doi.org/10.1161/circulationaha.111. 088633
- Quezada E, Lapidus J, Shaughnessy R, Chen Z, Silberbach M (2015) Aortic dimensions in Turner syndrome. Am J Med Genet A 167(11):2527–2532. https://doi.org/10.1002/ajmg.a.37208
- 107. Mortensen KH, Hjerrild BE, Stochholm K, Andersen NH, Sørensen KE, Lundorf E, Hørlyck A, Pedersen EM, Christiansen JS, Gravholt CH (2011) Dilation of the ascending aorta in Turner syndrome—a prospective cardiovascular magnetic resonance study. J Cardiovasc Magn Reson 13(1):24. https://doi.org/ 10.1186/1532-429x-13-24
- Carlson M, Silberbach M (2007) Dissection of the aorta in Turner syndrome: two cases and review of 85 cases in the literature. J Med Genet 44(12):745–749. https://doi.org/10.1136/jmg.2007. 052019
- 109. Baumgartner H, De Backer J, Babu-Narayan SV, Budts W, Chessa M, Diller GP, Lung B, Kluin J, Lang IM, Meijboom F, Moons P, Mulder BJM, Oechslin E, Roos-Hesselink JW, Schwerzmann M, Sondergaard L, Zeppenfeld K (2021) 2020 ESC guidelines for the management of adult congenital heart disease. Eur Heart J 42(6):563–645. https://doi.org/10.1093/eurheartj/ehaa5 54
- Mortensen KH, Erlandsen M, Andersen NH, Gravholt CH (2013) Prediction of aortic dilation in Turner syndrome-the use of serial cardiovascular magnetic resonance. J Cardiovasc Magn Reson 15(1):47. https://doi.org/10.1186/1532-429x-15-47
- 111. Mortensen KH, Cleemann L, Hjerrild BE, Nexo E, Locht H, Jeppesen EM, Gravholt CH (2009) Increased prevalence of autoimmunity in Turner syndrome–influence of age. Clin Exp Immunol 156(2):205–210. https://doi.org/10.1111/j.1365-2249. 2009.03895.x
- 112. Jørgensen KT, Rostgaard K, Bache I, Biggar RJ, Nielsen NM, Tommerup N, Frisch M (2010) Autoimmune diseases in women with Turner's syndrome. Arthritis Rheum 62(3):658–666. https:// doi.org/10.1002/art.27270
- 113. Bakalov VK, Gutin L, Cheng CM, Zhou J, Sheth P, Shah K, Arepalli S, Vanderhoof V, Nelson LM, Bondy CA (2012) Autoimmune disorders in women with Turner syndrome and women with karyotypically normal primary ovarian insufficiency. J Autoimmun 38(4):315–321. https://doi.org/10.1016/j.jaut.2012. 01.015
- 114. Sas TC, de Muinck Keizer-Schrama SM, Stijnen T, Aanstoot HJ, Drop SL (2000) Carbohydrate metabolism during long-term growth hormone (GH) treatment and after discontinuation of GH treatment in girls with Turner syndrome participating in a randomized dose-response study. Dutch advisory group on growth hormone. J Clin Endocrinol Metab 85(2):769–775. https://doi. org/10.1210/jcem.85.2.6334
- 115. Elsheikh M, Hodgson HJ, Wass JA, Conway GS (2001) Hormone replacement therapy may improve hepatic function in women with Turner's syndrome. Clin Endocrinol (Oxf) 55(2):227–231. https://doi.org/10.1046/j.1365-2265.2001.01321.x

116. Baronio F, Mazzanti L, Girtler Y, Tamburrino F, Lupi F, Longhi S, Fanolla A, Radetti G (2017) The influence of GH treatment on glucose homeostasis in girls with Turner syndrome: a 7-year study. J Clin Endocrinol Metab 102(3):878–883. https://doi.org/ 10.1210/jc.2016-3179

Authors and Affiliations

T. Aversa^{1,2} · L. De Sanctis^{3,4} · M. F. Faienza⁵ · A. Gambineri⁶ · A. Balducci⁷ · R. D'Aprile^{8,9} · C. Di Somma¹⁰ · C. Giavoli¹¹ · A. Grossi¹² · M. C. Meriggiola¹³ · E. Profka¹⁴ · M. Salerno¹⁵ · S. Stagi^{16,17} · E. Scarano¹⁸ · M. C. Zatelli¹⁹ · M. Wasniewska^{1,2}

- M. Wasniewska malgorzata.wasniewska@unime.it
- ¹ Department of Human Pathology of Adulthood and Childhood, University of Messina, Messina, Italy
- ² Pediatric Unit, University Hospital "G. Martino", Via Consolare Valeria N. 1, 98124 Messina, Italy
- ³ Pediatric Endocrinology, Regina Margherita Children Hospital, Turin, Italy
- ⁴ Department of Public Health and Pediatric Sciences, University of Turin, Turin, Italy
- ⁵ Department of Precision and Regenerative Medicine and Ionian Area, University of Bari "Aldo Moro", 70124 Bari, Italy
- ⁶ Division of Endocrinology and Diabetes Prevention and Care, Department of Medical and Surgical Sciences (DIMEC), IRCCS Azienda Ospedaliero – Universitaria di Bologna, Bologna, Italy
- ⁷ Pediatric Cardiology and Adult Congenital Heart Disease Program, Department of Cardio - Thoracic and Vascular Medicine, IRCCS Azienda Ospedaliero - Universitaria Di Bologna, Bologna, Italy
- ⁸ Department of Women's and Children's Health, University of Padua, Padua, Italy
- ⁹ A.Fa.D.O.C. Association OdV, Vicenza, Italy

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

- ¹⁰ Unit of Endocrinology, AOU Federico II, Naples, Italy
- ¹¹ Endocrinology Unit, Fondazione IRCCS Cà Granda Ospedale Maggiore Policlinico, Milan, Italy
- ¹² Endocrine Pathology of Chronic and Post-Tumor Diseases Unit, "Bambino Gesù" Pediatric Hospital, Rome, Italy
- ¹³ Division of Gynecology and Physiopathology of Reproduction, Department of Medical and Surgical Sciences (DIMEC), IRCCS Azienda Ospedaliero – Universitaria di Bologna, Bologna, Italy
- ¹⁴ Endocrinology Unit, Fondazione IRCCS Ca' Granda Ospedale Maggiore Policlinico, Milan, Italy
- ¹⁵ Pediatric Section, Department of Translational Medical Sciences, University of Naples Federico II, Naples, Italy
- ¹⁶ Health Sciences Department, University of Florence, Florence, Italy
- ¹⁷ Meyer Children's Hospital IRCCS, Florence, Italy
- ¹⁸ Pediatric Unit, IRCCS Azienda Ospedaliero Universitaria di Bologna, Bologna, Italy
- ¹⁹ Section of Endocrinology, Geriatrics and Internal Medicine, Department of Medical Sciences, University of Ferrara, Ferrara, Italy